



King's Research Portal

DOI: 10.1002/aur.1733

Document Version Peer reviewed version

Link to publication record in King's Research Portal

Citation for published version (APA):

Shephard, E., Milosavljevic, B., Pasco, G., Jones, E., Gliga, T., Happe, F. G. E., Johnson, M. H., & Charman, T. (2016). Mid-Childhood Outcomes of Infant Siblings at Familial High-Risk of Autism Spectrum Disorder. *Autism research*. https://doi.org/10.1002/aur.1733

Citing this paper

Please note that where the full-text provided on King's Research Portal is the Author Accepted Manuscript or Post-Print version this may differ from the final Published version. If citing, it is advised that you check and use the publisher's definitive version for pagination, volume/issue, and date of publication details. And where the final published version is provided on the Research Portal, if citing you are again advised to check the publisher's website for any subsequent corrections.

General rights

Copyright and moral rights for the publications made accessible in the Research Portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognize and abide by the legal requirements associated with these rights.

•Users may download and print one copy of any publication from the Research Portal for the purpose of private study or research. •You may not further distribute the material or use it for any profit-making activity or commercial gain •You may freely distribute the URL identifying the publication in the Research Portal

Take down policy

If you believe that this document breaches copyright please contact librarypure@kcl.ac.uk providing details, and we will remove access to the work immediately and investigate your claim.

Mid-Childhood Outcomes of Infant Siblings at Familial High-Risk of Autism Spectrum Disorder

Elizabeth Shephard^{a*}, Bosiljka Milosavljevic^b, Greg Pasco^b, Emily J. H. Jones^c, Teodora Gliga^c, Francesca Happé^a, Mark H. Johnson^c, Tony Charman^b and The BASIS Team^{**}

^a MRC Social, Genetic & Developmental Psychiatry Centre, Institute of Psychiatry,
Psychology, & Neuroscience, King's College London, De Crespigny Park, London, SE5 8AF
^b Department of Psychology, Henry Wellcome Building, Institute of Psychiatry, Psychology,
& Neuroscience, King's College London, De Crespigny Park, London, SE5 8AF.
^c Centre for Brain and Cognitive Development, School of Psychology, Birkbeck College,
Malet Street, London, WC1E 7HX

* Corresponding author. Email: elizabeth.1.shephard@kcl.ac.uk. Tel: +44 (0) 207 848 5272

** The BASIS Team (in alphabetical order): Simon Baron-Cohen, Rachael Bedford, Patrick Bolton, Susie Chandler, Mayada Elsabbagh, Janice Fernandes, Holly Garwood, Kristelle Hudry, Andrew Pickles, Leslie Tucker, Agnes Volein.

Running title: Mid-childhood Outcomes of HR Infant Siblings Main text word count = 5,599 Number of text pages = 18 Number of Tables = 3 Supplementary materials: Number of text pages = 11, Number of Tables = 4 Grant sponsor: UK Medical Research Council Grant number: G0701484; K021389

Scientific Abstract

Almost 20% of infants with an older sibling with autism spectrum disorder (ASD) exhibit ASD themselves by age 3 years. The longer-term outcomes of high-risk infants are less clear. We examined symptoms of ASD, attention-deficit/hyperactivity disorder (ADHD) and anxiety, language, IQ, and adaptive behaviour at age 7 years in high- and low-risk children prospectively studied since the first year of life. Clinical outcomes were compared between high-risk children who met diagnostic criteria for ASD at age 7 (HR-ASD-7 group, n = 15), high-risk children without ASD (HR-Non-ASD-7 group, n = 24), and low-risk control children (LR group, n = 37). Diagnostic stability between age 3 and 7 years was moderate, with five children who did not meet diagnostic criteria for ASD at age 3 years being assigned the diagnosis at age 7, and three children showing the opposite pattern. The HR-ASD-7 group showed elevated ADHD and anxiety symptoms and had lower adaptive behaviour scores than LR controls. The HR-Non-ASD-7 group had higher repetitive behaviour, lower adaptive functioning and elevated scores on one anxiety subscale (Separation Anxiety) compared to LR controls, but evidence for subclinical ASD symptoms (the broader autism phenotype, BAP) was limited in the group as a whole, although we identified a subgroup with elevated ASD traits. The difficulties experienced by high-risk siblings at school-age extend beyond ASD symptoms. The pattern of difficulties exhibited by the HR-ASD-7 group may inform our understanding of developmental trajectories of co-occurring psychopathology in ASD.

[Word count = 241]

Key words: High-risk siblings, Clinical outcomes, ADHD, Anxiety, Broader autism phenotype

Lay Abstract

Almost one-in-five infants at high familial risk for autism spectrum disorder (ASD), due to having an older sibling with an ASD diagnosis, develop ASD themselves by age 3 years. Less is known about the longer-term outcomes of high-risk infants. To address this issue, we examined symptoms of ASD and associated developmental conditions (attentiondeficit/hyperactivity disorder (ADHD); anxiety), language, IQ, and adaptive behaviour at age 7 years in high- and low-risk children studied from infancy. We compared outcomes between high-risk children who met criteria for ASD at age 7, high-risk children without ASD, and low-risk control children. Diagnostic stability between 3 and 7 years was moderate. Highrisk siblings with ASD showed elevated levels of ADHD and anxiety symptoms and lower adaptive behaviour than low-risk control children. High-risk siblings without ASD had higher repetitive behaviours, lower adaptive functioning, and elevated scores on one anxiety subscale (Separation Anxiety) compared to low-risk controls. The findings indicate that the difficulties experienced by high-risk siblings at school age extend beyond ASD symptoms. Better understanding of these difficulties may improve models of the development of cooccurring problems seen in children with ASD.

[Word count = 184]

Prospective studies of infants at familial high-risk of autism spectrum disorder (ASD) because of having an older sibling with ASD are predicated on the known elevated recurrence rate of ~10% (Constantino et al., 2010; Sandin et al., 2014) compared to the general population prevalence of ~1% (Baird et al., 2006; CDC, 2014). In prospective high-risk sibling studies the recurrence rates have been higher, with close to 20% of high-risk infants meeting diagnostic criteria for ASD when assessed at age 3 years (Messinger et al., 2015; Ozonoff et al., 2011). In addition, a further ~20% of non-ASD high-risk siblings show elevated ASD severity scores and/or lower levels of developmental ability when assessed at the same age (Charman et al., in press; Messinger et al., 2013).

Ozonoff et al. (2014) found that around a quarter of high-risk siblings who did not have ASD were 'non-typically developing' at age 3 based on scoring above threshold on the Autism Diagnostic Observational Schedule (ADOS; Lord et al., 2000) and/or low scores on the Mullen Scales of Early Learning (MSEL; Mullen, 1995). Clinical judgement of these children found that around one third showed aspects of the 'broader autism phenotype' (BAP) – subclinical social-communication and restricted, repetitive behaviour atypicalities seen at an elevated rate in family members of individuals with ASD (Bolton et al., 1994; Pickles et al., 2000; Piven et al., 1997) – while others showed features of emergent attentiondeficit/hyperactivity disorder (ADHD), general developmental delay, or speech and language difficulties (Ozonoff et al., 2014). In an overlapping sample, Miller et al. (2015a) reported elevated rates of pragmatic language difficulties and Schwichtenberg et al. (2013) slightly elevated anxiety, depression and aggression scores in high-risk non-ASD siblings at 3 years. The emerging pattern from these studies suggests that, from a young age, high-risk children are vulnerable to developing not only ASD and subclinical ASD traits, but also problems with language (Pickles et al., 2014; Wodka et al., 2013), ADHD and anxiety (Simonoff et al., 2008; White et al., 2009) that frequently co-occur with ASD and overlap at the genetic level (Hallett et al., 2013; Ronald et al., 2008; Tick et al., 2016).

Recently, groups have reported on clinical outcomes in high-risk siblings followed further, into the school-age years. Gamliel et al. (2009) reported that 41% of high-risk siblings without ASD showed either low scores on IQ, language or attainment measures and/or parent reported problems in social-communication and stereotyped behaviours at age 7 years. Subsequently, Ben-Yizhak et al. (2011) reported this subgroup of high-risk siblings to display elevated 'semantic-pragmatic' language difficulties compared to typically developing high- and low-risk siblings. In contrast, Gillespie-Lynch et al. (2015) reported difficulties with structural and pragmatic aspects of language development at age 7 only in high-risk siblings who themselves had ASD, and not in those without ASD, compared to low-risk controls. Similarly, Drumm et al. (2015) found that non-ASD high-risk siblings showed average/above-average expressive, receptive and pragmatic language abilities at 8-11 years, although pragmatic language was relatively weaker than structural language, and the siblings showed impairments in phonological aspects of language. Warren et al. (2012) reported on outcomes at age 5 years in non-ASD high-risk siblings and found few differences compared to low-risk controls on global measures of IQ, language and emotional and behavioural problems, but slightly elevated social cognition difficulties and restricted and repetitive behaviours. Miller et al. (2015b) reported higher internalising and externalising problems and poorer receptive and expressive language abilities in a group of high-risk non-ASD siblings with clinical concerns (BAP, ADHD-like problems, emotional problems, developmental delay) compared to typically developing high- and low-risk siblings, who did not differ from one another, at 5 to 9 years.

In summary, school-age follow-up studies indicate that ASD traits and language, behavioural and emotional problems persist into mid-childhood for some high-risk siblings

who do not develop ASD. However, there is some inconsistency in which domains are affected, perhaps reflecting heterogeneity in the development of high-risk non-ASD infants, or the restricted range of clinical outcomes (e.g. only language) examined in many studies. Another issue is that high-risk infants who do develop ASD have largely been excluded from analyses in school-age follow-up studies (e.g. Drumm et al., 2015; Miller et al., 2015b; Warren et al., 2012). This precludes the comparison of levels of co-occurring difficulties such as ADHD, anxiety and language problems between high-risk siblings who develop ASD and those who do not.

To date, only one study has reported on diagnostic stability of ASD into the schoolage years in high-risk children. Brian et al. (2015) reported that 17/18 (94.4%) high-risk toddlers with ASD continued to meet diagnostic criteria for ASD at 7-12 years while one child no longer met ASD criteria, and 6/49 (12.2%) non-ASD high-risk toddlers were classified as "Later-diagnosed" due to meeting diagnostic criteria for ASD in mid-childhood but not at age 3 years (*pace* DSM-5, which allows symptoms to date from 'the developmental period'). However, Brian et al. (2015) did not report on other developmental outcomes, and so the question of what other difficulties, such as co-occurring ADHD and anxiety, high-risk children with ASD experience remains to be answered.

The current study reports on a wide range of mid-childhood clinical outcomes as well as the stability of ASD diagnosis between age 3 and 7 years in a cohort of high-risk children, including both those who met diagnostic criteria for ASD at age 7 years (HR-ASD-7) and those who did not (HR-Non-ASD-7). Intelligence, language ability, adaptive behaviour and symptoms of ADHD and anxiety were compared between the two high-risk subgroups and a group of low-risk (LR) children to assess differences in mid-childhood clinical outcomes. Given sex differences in ASD symptomatology and developmental ability recently reported in high-risk siblings in early childhood, with boys more severely affected than girls (Messinger

et al., 2015), and known sex differences in rates of ADHD (males > females) and anxiety (females > males) in non high-risk samples (Arnett, Pennington, Willcutt, DeFries, & Olson, 2015; Lewinsohn, Gotlib, Lewinsohn, Seeley, & Allen, 1998), we explored whether sex influenced school-age outcomes. We predicted the following pattern of findings:

- The HR-ASD-7 group would show lower IQ, language ability and adaptive functioning, and higher symptoms of ADHD and anxiety compared to the HR-Non-ASD-7 group and LR controls.
- In line with the BAP concept, the HR-Non-ASD-7 group would show elevated ASD traits compared to the LR controls. These children may also show lower levels of IQ, language and adaptive behaviour and higher levels of ADHD and anxious traits than the LR controls.
- 3. Boys would show higher ASD and ADHD symptomatology than girls, while girls would show higher levels of anxiety than boys.

Methods

Participants

Participants were 104 children taking part in a prospective longitudinal study of infants at high- and low- familial risk for autism (hereafter, HR and LR) recruited as part of the British Autism Study of Infant Siblings (BASIS; www.basisnetwork.org). Siblings completed research visits at 7 and 14 months of age, around their second and third birthdays, and were invited to return for the current follow-up study at age 6-8 years. At enrolment, each HR infant (n = 54) had an older sibling (in 4 cases, a half-sibling) with a community clinical ASD diagnosis, confirmed using information from the *Development and Well-Being Assessment* (*DAWBA*; Goodman et al., 2000) and the *Social Communication Questionnaire* (*SCQ*; Rutter et al., 2003) by expert clinicians (TC, PB)¹. Parent-reported family medical

¹ 5 DAWBA and 5 SCQ missing

histories were examined for significant conditions in the proband or extended family members (e.g., Fragile X syndrome, tuberous sclerosis) with no such conditions reported. LR controls (n = 50) were full-term infants (with one exception) recruited from a volunteer database at the Birkbeck Centre for Brain and Cognitive Development. Medical history review confirmed lack of ASD within first-degree relatives. At enrolment, all LR infants had at least one older sibling. The SCQ was used to confirm absence of ASD in these older siblings, with no child scoring above instrument cut-off (≥ 15 ; n = 1 missing data).

Of 53 HR and 48 LR children retained at the 3 year assessment, 44 HR (83%) and 37 LR (77%) agreed to take part in the follow-up study at 6-8 years. Of these, two HR children did not complete a research visit (parents completed questionnaires only) and we were unable to assign them to an ASD outcome group and consequently excluded them from analysis, leaving a final sample of 42 HR siblings (15 boys, 27 girls) and 37 LR controls (15 boys, 22 girls). The HR and LR groups did not differ in age (HR mean (SD): 90.6 (6.3) months; LR mean (SD): 89.3 (4.9) months; t (75) = -.96, p = .34) or sex (HR % male: 35.7; LR % male: 40.5; χ^2 (1) = .20, p = .66) at the follow-up. The retained sample did not differ from the non-retained sample in 3 year levels of ASD on the ADOS, SRS, or SCQ, developmental level on the MSEL, adaptive behaviour assessed with the *Vineland Adaptive Behavior Scales – Second Edition* (*Vineland-II*; Sparrow et al., 2005), or family income (all p > .4). Ethical approval was obtained from the NHS National Research Ethics Service (NHS RES London REC 14/LO/0170). Parents provided written informed consent. Children provided written informed assent wherever possible given developmental level.

Outcome measures collected at 7 years

ASD Symptomatology

The *Autism Diagnostic Observation Schedule – Second Edition (ADOS-2*; Lord et al., 2012), a standardised interaction observation assessment, was used to assess current

symptoms of ASD (module 3 for 73 children, module 2 for one child, module 1 for one child; ADOS not completed with 3 LR children). Calibrated Severity Scores for Social Affect, Restricted and Repetitive Behaviours (RRB) and Overall Total were computed (Gotham et al., 2009; Hus et al., 2014), which provide standardised autism severity measures that account for differences in module administered, age and verbal ability. The *Autism Diagnostic Interview* – *Revised (ADI-R*; Le Couteur et al., 2003), a structured parent interview, was completed with parents of HR children. Standard algorithm scores were computed for Reciprocal Social Interaction (Social), Communication, and Restricted, Repetitive and Stereotyped Behaviours and Interests (RRB).

Parents completed questionnaires to obtain further information on ASD symptoms and subclinical ASD traits. The *Social Responsiveness Scale – Second Edition (SRS-2;* Constantino, 2012) assessed parent-rated social impairments over the past 6 months and ageand sex-normed T-scores (mean 50; SD 10) were used in analysis. The *Short Sensory Profile* (*SSP*; Dunn, 1999) assessed parent-rated sensory processing difficulties; the total SSP score was used in analysis. The *Repetitive Behavior Scale – Revised (RBS-R*; Bodfish et al., 1999) measured parent-rated repetitive/restricted behaviours; the total RBS-R score was used in analysis. Higher scores on the SRS-2 and RBS-R reflect more severe ASD symptoms; lower scores on the SSP reflect greater sensory processing problems.

IQ and adaptive behaviour

The *Wechsler Abbreviated Scale of Intelligence – Second Edition (WASI-II*; Wechsler, 2011), a standardised instrument to assess intellectual ability, was completed with each child. Age-normed intelligence quotients (mean 100; SD 15) for the verbal domain (Verbal Comprehension Index, VCI), performance domain (Perceptual Reasoning Index, PRI), and full-scale IQ (FSIQ) were used in analyses. One HR child was unable to complete the assessment due to intellectual disability. Adaptive behaviour was assessed using the Vineland-II (Sparrow et al., 2005), a semistructured interview in which parents rate their child's current level of functioning across the domains of Communication, Daily Living and Socialization. Age-normed Standard Scores (mean 100; SD 15) were computed for each domain and the Adaptive Behavior Composite (ABC).

Language

The *Clinical Evaluation of Language Fundamentals – Fourth Edition UK (CELF-4*; Semel et al., 2006) is an age-normed standardised instrument to assess language ability. The Concepts and Following Directions and Recalling Sentences subtests were completed with HR and LR children to assess receptive and expressive language, respectively (mean 10; SD 3). Two HR children had insufficient language to complete the assessment; a further four HR and 10 LR children did not complete the assessment due to time constraints. Standard Scores on the Communication domain of the Vineland-II were used as an additional measure of language ability.

ADHD and anxiety

The parent-rated *Conners 3* (Conners, 2008) was used to assess symptoms of ADHD over the past 6 months. T-scores (mean 50; SD 10) for the DSM-IV-TR Inattentive and Hyperactive/Impulsive scales were used in analyses. The parent-report *Spence Children's Anxiety Scale (SCAS*; Spence, 1998) was used to assess current symptoms of children's anxiety. The SCAS is based on DSM-IV-TR (American Psychiatric Association, 2000) criteria for anxiety disorders and generates scores for the domains of Separation Anxiety, Social Phobia, Obsessive Compulsive Disorder (OCD), Physical Injury Fears, Panic/Agoraphobia, and Generalized Anxiety, as well as a Total Anxiety score. Scores on the six anxiety domains and the total SCAS score were used in analyses. We were also interested in how many children in each group showed ADHD and anxiety scores in the clinically

significant range. We therefore calculated the percentage of children per group with T-scores > 65 on the Conners (T-scores over 65 are considered to be indicative of clinically significant symptoms, Conners, 2008) and the percentage of children with z-transformed SCAS Total Anxiety scores 1.5 SD above the mean of a large (n = 261) normative sample of typically developing 6-11 year olds (Nauta et al., 2004).

Assignment to ASD outcome groups at 3 and 7 years

3 years: Experienced researchers who conducted the assessments (KH, SC, GP) and the lead clinician (TC) reviewed information across the research visits at 2 years (including ADOS-G, MSEL and Vineland assessments) and 3 years (including MSEL, Vineland, ADOS-G and ADI-R) and as a team assigned clinical consensus best estimate diagnosis of ASD according to ICD-10 (World Health Organisation, 1993). For LR control children, in the absence of a full developmental history (no ADI-R was administered) no formal clinical clinical diagnoses were considered but none had a community clinical ASD diagnosis.

7 years: Experienced researchers who conducted the assessments (ES, BM, GP) and the lead clinician (TC) reviewed information on ASD symptomatology (ADOS-2, ADI-R (HR only), SCQ), adaptive functioning (Vineland-II), and IQ (WASI-II) for each HR and LR child and as a team assigned clinical consensus best estimate diagnosis of ASD according to DSM-5 (American Psychiatric Association, 2013). None of the 37 LR children met DSM-5 criteria for ASD and none had a community clinical ASD diagnosis.

Diagnosis at age 7 years included review of all information previously obtained, including from the 2 year and 3 year visits. Note that there was overlap in the personnel involved in the diagnostic decision-making (GP, TC) so the decisions were not independent. However, the age 7 diagnostic decisions were not directly 'yoked' to the diagnostic decisions previously taken at the 3 year visit but rather, in light of the large amount of additional information available about the children – in particular as by school age the ADI-R includes

more information (e.g. with respect to peer interactions; functioning outside the home setting) and uses the full set of diagnostic algorithm items which are not all applicable at 3 years – a decision was made as to whether the child currently met DSM-5 criteria for ASD. In line with the primary focus of the current paper on mid-childhood outcomes in HR siblings, and not diagnostic stability over time per se, we report findings in relation to 7 year ASD diagnosis made on the basis of information available from all clinical assessments up to that timepoint.

Statistical analysis

All statistical analyses were conducted in SPSS v21 (IBM Corp, 2012). The hypothesised group differences in 7 year outcomes were assessed using ANOVA or MANOVA tests with group (HR-ASD-7, HR-Non-ASD-7, LR) as a between-subjects factor. Measures with one dependent variable (SRS-2, RBS-R, SSP) were analysed using univariate ANOVAs. Measures with multiple dependent variables (ADOS-2, ADI-R, WASI-II, Vineland II, CELF, Conners 3, SCAS) were analysed using MANOVAs. Significant group effects were further investigated with planned contrasts between each pair of groups with Tukey's HSD correction applied to control for multiple comparisons. Sex differences in school-age outcomes and sex-by-group interactions were investigated using 2 (Group: LR, HR) x 2 (Sex: male, female) ANOVA/MANOVAs.

Following previous school-age follow-up studies (Gamliel et al., 2009; Miller et al., 2015b), we conducted supplementary analyses to explore differences in key outcome variables (ASD, IQ, adaptive behaviour, language, ADHD, anxiety) between HR non-ASD children who showed signs of atypical development, defined as scoring above clinical cutoffs for ASD on either the ADOS-2 or the ADI-R, and those with typical development (see Supplementary Materials).

Results

Diagnostic stability from 3 to 7 years

At age 7 years, 15 HR children (7 boys, 8 girls) met DSM-5 (APA, 2013) criteria for ASD and the remaining 27 children (8 boys, 19 girls) did not. Of the 13 HR siblings with an ASD diagnosis at 3 years also seen at 7 years, 10 retained their ASD diagnosis (76.9%) and 3 (23.1%) did not ('lost diagnosis' – See Discussion). Of the 29 HR who were not given an ASD diagnosis at 3 years also seen at 7 years, 24 did not meet diagnostic criteria for ASD (82.8%) at 7 years but 5 (17.2%) did meet ASD criteria ('later diagnosed'). Scores for the HR groups broken down by both 3-year and 7-year diagnostic outcome are shown in the Supplementary Materials.

Group characteristics and ASD symptomatology at 7 years

The 15 HR children meeting DSM-5 criteria for ASD at age 7, including the 5 'later diagnosed' children, formed the HR-ASD-7 group. Of the 27 HR children who did not meet ASD criteria at age 7, the 3 'lost diagnosis' children were excluded from further analysis (given they met ICD-10 criteria for ASD earlier in development), leaving 24 HR children in the HR-Non-ASD-7 group. Group characteristics, ASD symptomatology, and statistical test results are displayed in Table 1; Cohen's *d* effect sizes for pairwise group contrasts are given in Table S3 (Supplementary Materials). The groups did not differ in age or sex at the 7 year follow-up. As expected, the groups differed significantly on ADI-R, ADOS-2, SRS-2, RBS-R, and SSP measures of ASD symptoms. The HR-ASD-7 group had higher ADI-R scores than the HR-Non-ASD-7 group (all p < .001, all $d \ge 1.33$), higher ADOS-2 Social Affect, RRB, and Total CSS than the LR controls (all $p \le .001$, all $d \ge 2.02$), and higher SRS-2 and RBS-R and lower (more atypical) SSP scores than the HR-Non-ASD-7 group and LR controls (p = .004, d = .85). The HR-Non-ASD-7 group had higher ADOS-2 RRB CSS than the LR group (p = .003, d = .93).

[TABLE 1 HERE]

IQ, adaptive behaviour, and language at 7 years

WASI-II FSIQ differed significantly between groups, reflecting marginally lower scores in the HR-Non-ASD-7 group than the LR controls (p = .05, d = .75), although all groups' scores were in the typical range (Table 2). WASI-II VCI and PRI scores did not differ between groups. The groups differed significantly on all adaptive behaviour variables (Table 2). The HR-ASD-7 group had lower ABC, Socialization, and Daily Living scores than the HR-Non-ASD-7 and LR groups (all $p \le .005$, all $d \ge .85$). The HR-Non-ASD-7 group had lower ABC and Daily Living scores than the LR controls (all $p \le .04$, all $d \ge .77$). For language variables, the groups differed significantly on Vineland Communication, with lower scores in the HR-ASD-7 group than the LR group (p = .02, d = .81), but not in receptive or expressive performance on the CELF (Table 2).

[TABLE 2 HERE]

ADHD and anxiety at 7 years

The groups differed significantly in Conners Hyperactive/Impulsive and Inattentive Tscores (Table 3). The HR-ASD-7 group had higher scores on both ADHD domains than the LR controls (both $p \le .007$, both $d \ge .90$), while the remaining group pairs did not differ. MANOVA on SCAS anxiety scores revealed significant group differences in Total Anxiety and most subscales (Table 3). The HR-ASD-7 group had significantly higher scores than LR controls for Separation Anxiety, OCD, Panic/Agoraphobia, Generalized Anxiety, and Total Anxiety (all $p \le .02$, $d \ge .64$), and higher Panic/Agoraphobia scores than the HR-Non-ASD-7 group (p = .01, d = .66). The HR-Non-ASD-7 group had higher Separation Anxiety scores than the LR controls (p = .04, d = .77). The HR-ASD-7 group had the largest proportion of children scoring above clinical cut-offs for the Conners ADHD domains and SCAS Total Anxiety, followed by the HR-Non-ASD-7 group and then the LR group. Due to small numbers of LR children scoring above threshold, these proportional measures were not analysed statistically.

[TABLE 3 HERE]

Sex differences in outcome variables

There were significant effects of sex on ADOS-2 Total (F(1, 68) = 7.53, p = .008, $\eta^2 = .100$), Social Affect (F(1, 68) = 7.28, p = .009, $\eta^2 = .097$) and RRB (F(1, 68) = 4.38, p = .04, $\eta^2 = .060$) CSS, Vineland Communication (F(1, 68) = 6.51, p = .01, $\eta^2 = .087$), and Generalised Anxiety (F(1, 70) = 4.07, p = .05, $\eta^2 = .055$). Boys had significantly higher ADOS scores and lower Communication and Generalised Anxiety than girls across HR and LR groups. There was a significant group-by-sex interaction for FSIQ (F(1, 69) = 5.52, p = .03, $\eta^2 = .074$), reflecting a trend for higher scores in girls than boys in the LR group only.

Discussion

Stability of diagnosis

Stability of ASD diagnosis between 3 and 7 years was moderate. Ten HR children met criteria at both time-points ('Stable diagnosis'), 3 children who met criteria at age 3 did not at 7 ('Lost diagnosis'), and 5 children showed the opposite pattern ('Later diagnosed'). The 'Lost diagnosis' subgroup had high ADOS scores, moderate ADI-R scores and average developmental abilities at 3 years (Table S1), but by age 7 showed ADI-R and ADOS scores similar to the HR 'Never diagnosed' subgroup and a high-average mean IQ (FSIQ = 118) (Table S2). Two children who 'Lost diagnosis' were considered clinically to show traits of ASD at 7 years but did not reach the clinical threshold for ASD. The remaining 'Lost diagnosis' child was considered by the research team and parents to be typically developing at 7, and whilst parents had reported considerable concerns at age 3, they had no concerns at age 7. The 'Later diagnosed' subgroup had low ADI scores, ADOS scores similar to the HR 'Never diagnosed' subgroup, and high developmental and adaptive ability at age 3 (Table S1). At age 7, however, their ADI and ADOS scores were high and adaptive behaviour was low, similar to the 'Stable diagnosis' group, despite having high-average intelligence (Table S2). A similar pattern of findings has been reported for later-diagnosed high-risk children (Brian et al., 2015) and in clinical samples (Davidovitch, Levit-Binnun, Golan & Mannin-Courtney, 2015). The 'Later diagnosed' subgroup also had high ADHD scores and the highest anxiety scores at 7 (Table S2).

It is known from clinically referred samples that diagnostic stability from toddlerhood to mid-childhood of more narrowly defined autism is relatively good (e.g. Charman, Taylor, Drew, Cockerill, Brown & Baird, 2005; Lord, Risi, DiLavore, Shulman, Thurm & Pickles, 2006). However, for broader ASD stability from the preschool years to mid-childhood is only moderate (Rondeau et al., 2011; Woolfenden et al., 2012). Stability might be expected to be lower in familial high-risk samples compared to clinically referred samples where parents and/or professionals have explicit concerns about a child's development. Diagnostic decision-making is a complex process in which measurement error and some degree of uncertainty is to be expected (Constantino & Charman, 2016). In clinically referred cohorts, across the preschool to mid-childhood age-range levels of symptoms are relatively stable but there are some children whose symptoms lessen over time (termed 'improvers' in Gotham et al., 2012) and others whose symptoms worsen (see also Fountain, Winter & Bearman, 2012). In common with other behaviourally defined disorders (Tandon, Tillman, Agrawal & Luby, 2016) – alongside the aforementioned measurement error – some children will inevitably cross a diagnostic threshold or boundary as symptoms fluctuate over time. This is compounded in relation to ASD by the commonly co-occurring conditions, including ADHD and anxiety, which also wax-and-wane in severity over development. These factors may have added to diagnostic instability across the 3 and 7 year diagnostic assessments, notwithstanding the use of standard diagnostic instruments by an experienced team. Other

contributing factors include possible use of interventions for ASD (although none of the 'Lost diagnosis' children had received specific interventions) and changes in diagnostic criteria used for 3-year and 7-year diagnoses (ICD-10 at 3 and DSM-5 at 7). Given the small sample sizes of these diagnostic subgroups and the limitation in the current study that the diagnostic assessments at age 3 and 7 years were not independent our ability to interpret these findings is limited and will only be possible when larger samples of HR siblings are followed into mid-childhood.

Recurrence rate

It is important to note that the recurrence rate in our sample is higher than that of ~19% reported in the overlapping BSRC pooled samples (Ozonoff et al., 2011; Messinger et al., 2015). This was true at age 3 years (32%, Elsabbagh et al., 2012) and even more so at 7 years (36%), reflecting the fact that some children not diagnosed at age 3 years were considered to meet criteria at age 7. There is considerable cross-site variability in recurrence rates and other studies have recently reported similarly elevated recurrence rates at age 2 years (29%, Elison et al., 2014) and 3 years (28%, Zwaigenbaum et al., 2016). Our elevated recurrence rate might partially reflect the modest sample size (n = 53 at 3 years, n = 42 at 7 years) and whilst there was no *systematic* evidence of differences in retention from 3-to-7-years based on 3 year autism severity or developmental data it does limit the generalisability of our findings to high-risk siblings more generally.

Non ASD outcomes

As predicted, the HR-ASD-7 group had poorer outcomes than the other groups on several measures. Unimpaired IQ (FSIQ \geq 70) in all but two children² indicated this group mainly represented 'high-functioning' children with ASD. Nevertheless, these children showed significantly lower adaptive behaviour than the LR group, consistent with previously

² One HR-ASD-7 child could not complete the WASI-II due to intellectual impairment; a second HR-ASD-7 child had FSIQ < 70.

reported discrepancies between IQ ('capacity') and the ability to cope in everyday life ('competency') in ASD (Ashwood et al., 2015; Charman et al., 2011). In line with the high co-occurrence of ASD, anxiety and ADHD in population studies (Simonoff et al., 2008) and evidence for genetic overlap between these disorders (Hallett et al., 2013; Ronald et al., 2008; Tick et al., 2016), the HR-ASD-7 group had significantly higher ADHD and anxiety scores than LR children. Further, a large proportion of this group scored in the clinical range on the Hyperactive/Impulsive (53%) and Inattentive (33%) ADHD domains and the SCAS Total Anxiety (27%) measure. The pattern of findings in the HR-ASD-7 group demonstrates that these children experience a range of difficulties beyond ASD by early school-age and highlights the need to assess the longer-term and broader clinical outcomes of HR infants who develop ASD, which have only been reported in one previous study (Gillespie-Lynch et al., 2015). Our findings also have implications for understanding co-occurring psychopathology in ASD more generally. In non-high-risk ASD samples, anxiety disorders (~40%) and ADHD (~30%) are two of the most common co-occurring psychiatric conditions (Simonoff et al., 2008). Our findings indicate that similar rates of ADHD and anxiety may co-occur with ASD in non-clinically referred samples (see Tick et al., 2016 for similar findings in a population-based twin sample). This is important since comorbidities may be overrepresented in clinically-referred samples.

Contrary to our hypotheses, there was limited evidence of the BAP at the group level in the HR children without ASD. The HR-Non-ASD-7 group had average IQ and did not differ significantly from the LR group in social-communication abilities or language, although they did have higher RRB ADOS scores and lower adaptive behaviour. Warren et al. (2012) also found elevated RRB scores in their group of non-ASD HR siblings, suggesting such children might be particularly vulnerable to difficulties in this symptom domain. Other school-age follow-up studies have reported social-communication impairments in a

proportion of non-ASD HR children (Gamliel et al., 2009; Miller et al., 2015b), as have crosssectional studies of school-age non-ASD siblings (Constantino et al., 2006). Consistent with those studies, we did identify a sub-set of the HR-Non-ASD-7 siblings who showed subclinical ASD traits, defined as scoring above clinical cut-offs for ASD on either the ADOS or ADI-R but not meeting clinical criteria for ASD (HR-Atypical-7 group, see Supplementary Materials). We investigated whether this subgroup also showed problems with intellectual ability, language and adaptive behaviour, but, in contrast to our predictions, found no evidence for this. The HR-Atypical-7 group did not differ from LR controls or typically developing HR non-ASD siblings (HR-Typical-7 group) on FSIQ, ABC, or CELF language variables (Table S4), indicating they were functioning as well as their peers without ASD traits.

In terms of emotional and behavioural problems, as a group the HR-Non-ASD-7 children did not show raised ADHD symptoms compared to the LR controls. However, 20-30% of the HR-Non-ASD-7 group scored above the clinical threshold on the Conners indices, about double the rate in the LR controls, indicating there is a higher risk for inattention/hyperactivity difficulties in non-ASD HR children than LR children. The elevated Separation Anxiety scores and higher rates of anxiety scores in the clinical range in the HR-Non-ASD-7 group than in the LR controls indicates these children had difficulties with anxiety. This group also had (non-significantly) higher mean scores on Total and Generalised Anxiety than the LR controls (Table 2), which may have reached significance in a larger sample given the large effect sizes for these comparisons (Table S3). These findings are consistent with Miller et al. (2015b) who found that a proportion of non-ASD HR infants showed raised anxiety/depression or ADHD problems at school-age. Warren et al (2012) found no evidence of elevated internalising or externalising scores at school-age in non-ASD HR siblings, but they collapsed across symptom categories rather than examining

anxiety/internalising and ADHD/externalising domains separately. In family studies around 12% of non-ASD siblings in simplex families with a child with ASD have an ADHD diagnosis, rising to 18% in multiplex families (Oerlemans et al., 2015). Anxiety disorders have also been reported at higher rates in older family members of individuals with an ASD diagnosis than controls (Bolton et al., 1998; Piven & Palmer, 1999).

We examined whether the subgroup of non-ASD HR children with atypical outcomes showed raised ADHD and anxiety scores compared to the typically developing HR and LR children. In contrast to our predictions, we found that the HR-Atypical-7 group had comparable ADHD and anxiety scores to the HR-Typical-7 and LR groups (Table S4). This suggests that the proportion of HR non-ASD children scoring in the clinical range on the ADHD and anxiety measures does not correspond to the proportion of HR non-ASD siblings with elevated ASD traits. Examining early-life predictors of later ADHD and anxiety symptoms might help identify this subgroup of children in future. Note that our definition of 'atypical outcomes' in some HR children overlaps with but also differs from that adopted in previous studies. We used scores above the clinical cut-off on the ADOS-2 and ADI-R to identify this group but only in children who were clinically not judged to meet diagnostic criteria for ASD. This is consistent with some previous reports (e.g. Landa et al., 2007; 2013) but is stricter than in some studies where scores one point (e.g. Charman et al., 2016) or a few points (e.g. Ozonoff et al., 2014) below the clinical threshold have been used – although in all samples this includes children who scores extend into the clinical range. Unlike other groups (e.g. Ozonoff et al., 2014; Paul et al., 2011) we did not attempt to apply a clinical judgement about possible or likely 'BAP' presentations – in part because there are no agreed criteria for what these would be. If anything our stricter definition of 'atypicality' might have made it more likely that we would find elevated ADHD and anxiety scores in these children but this was not the case.

Taken together with previous mixed reports of spared and impaired cognitive ability, language, social-communication, and mood and behavioural problems in non-ASD HR siblings at school-age (Ben-Yizhak et al., 2011; Drumm et al., 2015; Gamliel et al., 2009; Gillespie-Lynch et al., 2015; Miller et al., 2015b; Warren et al., 2012), our findings highlight the heterogeneity in outcomes of at-risk children. It is clear from the mean IQ and Vineland scores of the HR-Non-ASD-7 group (see Table 2) that they are functioning in the average ability range and, at least in our modest size sample, there was limited evidence at a group level that they were experiencing significant developmental problems. It will now be important to establish what intrinsic and extrinsic factors influence whether a high-risk child without ASD will experience BAP characteristics or co-occurring emotional and behavioural problems.

We examined whether sex of the HR sibling affected outcomes. In line with recent findings (Messinger et al., 2015) we found that boys had higher ADOS scores, lower adaptive communicative behaviour and lower levels of Generalised Anxiety than girls. In the LR group only, girls had higher FSIQ scores than boys. There were no sex-by-group interactions on measures of ASD severity or behavioural outcomes indicating no strong sex specific liability to elevated levels of ASD, ADHD and anxiety symptoms in HR siblings (see also Zwaigenbaum et al., 2012), although our modest sample sized means we were under-powered to detect such differences.

Limitations

Our sample sizes for the high-risk subgroups were modest and this should be considered when interpreting our findings. The recurrence rate in our sample is high compared to the 3-year estimates from the large pooled BSRC dataset which may limit the generalisability of findings to high-risk siblings more broadly. Further, although there are many strengths to our design, including the use of well-established and validated instruments

to measure ASD and other phenotypic characteristics, there are also limitations in the implementation of the design, including a lack of independence between the 3 year and 7 year diagnostic judgements. In common with other high-risk sibling studies our HR children – including those with ASD – have high IQs compared to clinical samples of children with ASD, which may also limit generalisability. However, the clinical relevance of our findings is supported by the large effect sizes of significant group differences. Finally, although we carefully assessed ASD symptomatology with observer- and parent-rated measures we were unable to conduct similarly in-depth assessments of ADHD and anxiety and relied upon parent-report questionnaires. These questionnaires are well-validated and relate to DSM criteria for ADHD and anxiety disorders, but future work should undertake multi-informant assessments to obtain a more reliable characterisation of symptoms and co-occurring diagnoses.

Acknowledgements

We are very grateful for the important contributions BASIS families have made towards this study. The research was supported by the BASIS funding consortium led by Autistica (www.basisnetwork.org), Autism Speaks, UK Medical Research Council Programme Grants (G0701484 and K021389), as well as support from the Innovative Medicines Initiative Joint Undertaking under grant agreement n° 115300, resources of which are composed of financial contribution from the European Union's Seventh Framework Programme (FP7/2007 - 2013) and EFPIA companies' in kind contribution. The authors have no conflicts of interests.

The BASIS Team (in alphabetical order): Simon Baron-Cohen, Rachael Bedford, Patrick Bolton, Susie Chandler, Mayada Elsabbagh, Janice Fernandes, Holly Garwood, Kristelle Hudry, Andrew Pickles, Leslie Tucker, Agnes Volein.

References

- American Psychiatric Association. (2000). *Diagnostic and Statistical Manual of Mental Disorders* 4th Edition, text revision (DSM-IV-TR). Washington DC: American Psychiatric Association.
- American Psychiatric Association. (2013). *Diagnostic and Statistical Manual of Mental Disorders* (*DSM-5*). Washington DC: American Psychiatric Association.
- Arnett, A. B., Pennington, B. F., Willcutt, E. G., DeFries, J. C., & Olson, R. K. (2015). Sex differences in ADHD symptom severity. *Journal of Child Psychology and Psychiatry*, 56(6), 632-639.
- Ashwood, K. L., Tye, C., Azadi, B., Cartwright, S., Asherson, P., & Bolton, P. (2015). Brief
 Report: Adaptive functioning in children with ASD, ADHD and ASD + ADHD. *Journal of Autism and Developmental Disorders*, 45(7), 2235–2242.
 http://doi.org/10.1007/s10803-014-2352-y
- Baird, G., Simonoff, E., Pickles, A., Chandler, S., Loucas, T., Meldrum, D., & Charman, T.
 (2006). Prevalence of disorders of the autism spectrum in a population cohort of children in
 South Thames: the Special Needs and Autism Project (SNAP). *The Lancet*, *368*(9531), 210–215. http://doi.org/10.1016/S0140-6736(06)69041-7
- Ben-Yizhak, N., Yirmiya, N., Seidman, I., Alon, R., Lord, C., & Sigman, M. (2010). Pragmatic language and school related linguistic abilities in siblings of children with autism. *Journal of Autism and Developmental Disorders*, 41(6), 750–760. http://doi.org/10.1007/s10803-010-1096-6
- Bodfish, J., Symons, F., Lewis, M. (1999). *The Repetitive Behavior Scale*. Western Carolina Research Reports.
- Bolton, P., Macdonald, H., Pickles, A., Rios, P., Goode, S., Crowson, M., ... Rutter, M. (1994). A case-control family history study of autism. *Journal of Child Psychology and Psychiatry*, 35(5), 877–900. http://doi.org/10.1111/j.1469-7610.1994.tb02300.x

- Bolton, P. F., Pickles, A., Murphy, M., & Rutter M. (1998). Autism, affective and other psychiatric disorders: patterns of familial aggregation. *Psychological Medicine*, *28*(*2*), 385-395.
- Brian, J., Bryson, S.E., Smith, I.M., Roberts, W., Roncadin, C., Szatmari, R., & Zwaigenbaum, L. (2015). Stability and change in autism spectrum disorder diagnosis from age 3 to middle childhood in a high-risk sibling cohort. *Autism, published online ahead of print Dec 15*. doi:10.1177/1362361315614979.
- Developmental Disabilities Monitoring Network Surveillance Year 2010 Principal Investigators, & Centers for Disease Control and Prevention (CDC). (2014). Prevalence of autism spectrum disorder among children aged 8 years - autism and developmental disabilities monitoring network, 11 sites, United States, 2010. *Morbidity and Mortality Weekly Report. Surveillance Summaries (Washington, D.C.: 2002)*, 63(2), 1–21.
- Charman, T., Pickles, A., Simonoff, E., Chandler, S., Loucas, T., & Baird, G. (2011). IQ in children with autism spectrum disorders: data from the Special Needs and Autism Project (SNAP). *Psychological Medicine*, *41*(03), 619–627. http://doi.org/10.1017/S0033291710000991
- Charman, T., Taylor, E., Drew, A., Cockerill, H., Brown, J. A., & Baird, G. (2005). Outcome at 7 years of children diagnosed with autism at age 2: Predictive validity of assessments conducted at 2 and 3 years of age and pattern of symptom change over time. *Journal of Child Psychology and Psychiatry*, 46(5), 500-513.
- Charman, T., Young, G.S., Brian, J.,, Zwaigenbaum, L. (2016). Non-ASD outcomes at 36 months in siblings at familial risk for autism spectrum disorder.: A Baby Siblings Research Consortium (BSRC) study. *Autism Research*, doi: 10.1002/aur.1669
- Conners, K. (2008). Conners 3rd Edition (Conners 3). Toronto, Canada: Multi Health Systems.
- Constantino, J.N. (2012). Social Responsiveness Scale, Second Edition (SRS-2). Los Angeles, CA: Western Psychological Services.

- Constantino, J. N., & Charman, T. (2016). Diagnosis of autism spectrum disorder: reconciling the syndrome, its diverse origins, and variation in expression. *The Lancet Neurology*, *15*(3):279-91. doi: 10.1016/S1474-4422(15)00151-9.
- Constantino, J. N., Lajonchere, C., Lutz, M.,..., Todd, R. D. (2006). Autistic social impairment in the siblings of children with pervasive developmental disorders. *American Journal of Psychiatry*, 163(2), 294-296.
- Constantino, J. N., Zhang, Y., Frazier, T., Abbacchi, A. M., & Law, P. (2010). Sibling recurrence and the genetic epidemiology of autism. *The American Journal of Psychiatry*, *167*(11), 1349– 1356. http://doi.org/10.1176/appi.ajp.2010.09101470
- Davidovitch, M., Levit-Binnun, N., Golan, D., & Manning-Courtney, P. (2015). Late diagnosis of autism spectrum disorder after initial negative assessment by a multidisciplinary team. *Journal of Developmental & Behavioral Pediatrics*, 36(4), 227-234.
- Drumm, E., Bryson, S., Zwaigenbaum, L., & Brian, J. (2015). Language-related abilities in 'unaffected' school-aged siblings of children with ASD. *Research in Autism Spectrum Disorders*, 18, 83–96. http://doi.org/10.1016/j.rasd.2015.07.007

Dunn, W. (1999). Short sensory profile. San Antonio, TX: Psychological Corporation.

- Elison, J. T., Wolff, J. J., Reznick, J. S., Botteron, K. N., Estes, A. M., Gu, H., ... & Piven, J.
 (2014). Repetitive behavior in 12-month-olds later classified with autism spectrum disorder. *Journal of the American Academy of Child & Adolescent Psychiatry*, 53(11), 1216-1224.
- Elsabbagh, M., Mercure, E., Hudry, K., Chandler, S., Pasco, G., Charman, T., ... & BASIS Team. (2012). Infant neural sensitivity to dynamic eye gaze is associated with later emerging autism. *Current Biology*, 22(4), 338-342.
- Fountain, C., Winter, A. S., & Bearman, P. S. (2012). Six developmental trajectories characterize children with autism. *Pediatrics*, *129*(5), e1112-e1120.

- Gamliel, I., Yirmiya, N., Jaffe, D. H., Manor, O., & Sigman, M. (2009). Developmental trajectories in siblings of children with autism: Cognition and language from 4 months to 7 years. *Journal of Autism and Developmental Disorders*, 39(8), 1131–1144. http://doi.org/10.1007/s10803-009-0727-2
- Gillespie-Lynch, K., Khalulyan, A., del Rosario, M., McCarthy, B., Gomez, L., Sigman, M., & Hutman, T. (2015). Is early joint attention associated with school-age pragmatic language? *Autism*, 19(2), 168–177. http://doi.org/10.1177/1362361313515094
- Goodman, R., Ford, T., Richards, H., Gatward, R., & Meltzer, H. (2000). The Development and Well-Being Assessment: Description and initial validation of an integrated assessment of child and adolescent psychopathology. *The Journal of Child Psychology and Psychiatry and Allied Disciplines*, 41(05), 645–655.
- Gotham, K., Pickles, A., & Lord, C. (2008). Standardizing ADOS scores for a measure of severity in autism spectrum disorders. *Journal of Autism and Developmental Disorders*, *39*(5), 693– 705. http://doi.org/10.1007/s10803-008-0674-3
- Gotham, K., Pickles, A., & Lord, C. (2012). Trajectories of autism severity in children using standardized ADOS scores. *Pediatrics*, 130(5), e1278-e1284.
- Hallett, V., Lecavalier, L., Sukhodolsky, D. G., Cipriano, N., Aman, M. G., McCracken, J. T., ...
 Scahill, L. (2013). Exploring the manifestations of anxiety in children with autism spectrum disorders. *Journal of Autism and Developmental Disorders*, 43(10), 2341–2352.
 http://doi.org/10.1007/s10803-013-1775-1
- Hus, V., Gotham, K., & Lord, C. (2012). Standardizing ADOS domain scores: Separating severity of social affect and restricted and repetitive behaviors. *Journal of Autism and Developmental Disorders*, 44(10), 2400–2412. <u>http://doi.org/10.1007/s10803-012-1719-1</u>

- Landa, R. J., Gross, A. L., Stuart, E. A., & Faherty, A. (2013). Developmental trajectories in children with and without autism spectrum disorders: the first 3 years. *Child development*, 84(2), 429-442.
- Landa, R. J., Holman, K. C., & Garrett-Mayer, E. (2007). Social and communication development in toddlers with early and later diagnosis of autism spectrum disorders. *Archives of general psychiatry*, 64(7), 853-864.
- Le Couteur, A., Lord, C., & Rutter, M. (2003). *The Autism Diagnostic Interview-Revised (ADI-R)*. Los Angeles, CA: Western Psychological Services.
- Lewinsohn, P. M., Gotlib, I. H., Lewinsohn, M., Seeley, J. R., & Allen, N. B. (1998). Gender differences in anxiety disorders and anxiety symptoms in adolescents. *Journal of abnormal psychology*, *107*(1), 109.
- Lord, C., Risi, S., Lambrecht, L., Jr, E. H. C., Leventhal, B. L., DiLavore, P. C., ... Rutter, M. (2000). The Autism Diagnostic Observation Schedule—Generic: A standard measure of social and communication deficits associated with the spectrum of autism. *Journal of Autism and Developmental Disorders*, *30*(3), 205–223. http://doi.org/10.1023/A:1005592401947
- Lord, C., Risi, S., DiLavore, P. S., Shulman, C., Thurm, A., & Pickles, A. (2006). Autism from 2 to 9 years of age. *Archives of general psychiatry*,63(6), 694-701.
- Lord, C., Rutter, M., DiLavore, P. C., Risi, S., Gotham, K., & Bishop, S. L. (2012). Autism Diagnostic Observation Schedule: ADOS-2. Los Angeles, CA: Western Psychological Services.
- Messinger, D., Young, G. S., Ozonoff, S., Dobkins, K., Carter, A., Zwaigenbaum, L., . . . Sigman, M. (2013). Beyond autism: A Baby Siblings Research Consortium Study of high-risk children at three years of age. *Journal of the American Academy of Child and Adolescent Psychiatry*, 52(3), 300-308. doi: 10.1016/j.jaac.2012.12.011

- Messinger, D. S., Young, G. S., Webb, S. J., Ozonoff, S., Bryson, S. E., Carter, A., ...
 Zwaigenbaum, L. (2015). Early sex differences are not autism-specific: A Baby Siblings
 Research Consortium (BSRC) study. *Molecular Autism*, 6(1). http://doi.org/10.1186/s13229-015-0027-y
- Miller, M., Iosif, A.-M., Young, G. S., Hill, M., Phelps Hanzel, E., Hutman, T., ... Ozonoff, S. (2015b). School-age outcomes of infants at risk for autism spectrum disorder. *Autism Research*, doi.org/10.1002/aur.1572
- Miller, M., Young, G. S., Hutman, T., Johnson, S., Schwichtenberg, A. j., & Ozonoff, S. (2015a).
 Early pragmatic language difficulties in siblings of children with autism: implications for
 DSM-5 social communication disorder? *Journal of Child Psychology and Psychiatry*, 56(7),
 774–781. http://doi.org/10.1111/jcpp.12342
- Mullen, E. M. (1995). *Mullen scales of early learning*. Circle Pines, MN: American Guidance Service.
- Nauta, M.H., Scholing, A., Rapee, R.M., Abbott, M., Spence, S.H., & Waters, A. (2004). A parentreport measure of children's anxiety: psychometric properties and comparison with childreport in a clinic and normal sample. *Behaviour Research and Therapy*, *42*, 813-839.
- Oerlemans, A. M., Hartman, C. A., De Bruijn, Y. G., Van Steijn, D. J., Franke, B., Buitelaar, J. K., & Rommelse, N. N. (2015). Simplex and Multiplex Stratification in ASD and ADHD
 Families: A Promising Approach for Identifying Overlapping and Unique Underpinnings of ASD and ADHD? *Journal of autism and developmental disorders*, 45(3), 645-657.
- Ozonoff, S., Young, G. S., Belding, A., Hill, M., Hill, A., Hutman, T., ... Iosif, A.-M. (2014). The broader autism phenotype in infancy: When does It emerge? *Journal of the American Academy of Child & Adolescent Psychiatry*, 53(4), 398–407.e2.
 http://doi.org/10.1016/j.jaac.2013.12.020

- Ozonoff, S., Young, G. S., Carter, A., Messinger, D., Yirmiya, N., Zwaigenbaum, L., ... Stone, W.
 L. (2011). Recurrence risk for autism spectrum disorders: A Baby Siblings Research
 Consortium study. *Pediatrics*, *128*(3), e488–e495. http://doi.org/10.1542/peds.2010-2825
- Ozonoff, S., Young, G. S., Landa, R. J., Brian, J., Bryson, S., Charman, T., ... & Zwaigenbaum, L. (2015). Diagnostic stability in young children at risk for autism spectrum disorder: a baby siblings research consortium study. *Journal of Child Psychology and Psychiatry*, 56(9), 988-998. doi:10.1111/jcpp.12421
- Paul, R., Fuerst, Y., Ramsay, G., Chawarska, K., & Klin, A. (2011). Out of the mouths of babes:
 Vocal production in infant siblings of children with ASD. *Journal of Child Psychology and Psychiatry*, 52(5), 588-598.
- Pickles, A., Anderson, D. K., & Lord, C. (2014). Heterogeneity and plasticity in the development of language: a 17-year follow-up of children referred early for possible autism. *Journal of Child Psychology and Psychiatry*, 55(12), 1354–1362. http://doi.org/10.1111/jcpp.12269
- Pickles, A., Starr, E., Kazak, S., Bolton, P., Papanikolaou, K., Bailey, A., ... Rutter, M. (2000).
 Variable expression of the autism broader phenotype: Findings from extended pedigrees. *The Journal of Child Psychology and Psychiatry and Allied Disciplines*, 41(04), 491–502.
- Piven, J., Palmer, P., Jacobi, D., Childress, D., & Arndt, S. (1997). Broader autism phenotype: evidence from a family history study of multiple-incidence autism families. *The American Journal of Psychiatry*, 154(2), 185–190.
- Piven, J., & Palmer, P. (1999). Psychiatric disorder and the broad autism phenotype: evidence from a family study of multiple-incidence autism families. *American Journal of Psychiatry*, 156(4), 557-563.
- Ronald, A., Simonoff, E., Kuntsi, J., Asherson, P., & Plomin, R. (2008). Evidence for overlapping genetic influences on autistic and ADHD behaviours in a community twin sample. *Journal of*

Child Psychology and Psychiatry, *49*(5), 535–542. http://doi.org/10.1111/j.1469-7610.2007.01857.x

- Rondeau, E., Klein, L. S., Masse, A., Bodeau, N., Cohen, D., & Guilé, J. M. (2011). Is pervasive developmental disorder not otherwise specified less stable than autistic disorder? A metaanalysis. *Journal of Autism and Developmental Disorders*, 41(9), 1267-1276.
- Rutter, M., Bailey, A., & Lord, C. (2003). *The social communication questionnaire: Manual*. Los Angeles, CA: Western Psychological Services.
- Sandin S, Lichtenstein P, Kuja-Halkola R, Larsson H, Hultman CM, & Reichenberg A. (2014). The familial risk of autism. *JAMA*, *311*(17), 1770–1777. http://doi.org/10.1001/jama.2014.4144
- Schwichtenberg, A. J., Young, G. S., Hutman, T., Iosif, A.-M., Sigman, M., Rogers, S. J., &
 Ozonoff, S. (2013). Behavior and sleep problems in children with a family history of autism. *Autism Research*, 6(3), 169–176. http://doi.org/10.1002/aur.1278
- Semel, E., Wiig, E., Secord, W. (2006). Clinical Evaluation of Language Fundamentals 4th Edition. San Antonio: Harcourt Assessment, Inc.
- Simonoff, E., Pickles, A., Charman, T., Chandler, S., Loucas, T., & Baird, G. (2008). Psychiatric disorders in children with autism spectrum disorders: Prevalence, comorbidity, and associated factors in a population-derived sample. *Journal of the American Academy of Child & Adolescent Psychiatry*, 47(8), 921–929. http://doi.org/10.1097/CHI.0b013e318179964f
- Sparrow, S. S., Balla, D. A., & Cicchetti, D. V. (2005). Vineland Adaptive Behavior Scales, Second Edition (Vineland-II). Bloomington, NM: Pearson Assessments.
- Spence, S. H. (1998). A measure of anxiety symptoms among children. *Behaviour Research and Therapy*, *36*(5), 545–566. <u>http://doi.org/10.1016/S0005-7967(98)00034-5</u>

- Tandon, M., Tillman, R., Agrawal, A., & Luby, J. (2016). Trajectories of ADHD severity over 10 years from childhood into adulthood. *ADHD Attention Deficit and Hyperactivity Disorders*, 1-10.
- Tick, B., Colvert, E., McEwen, F., Stewart, C., Woodhouse, E., Gillan, N., ... & Ronald, A. (2016). Autism Spectrum Disorders and other mental health problems: exploring etiological overlaps and phenotypic causal associations. *Journal of the American Academy of Child & Adolescent Psychiatry*, 55(2), 106-113.
- Warren, Z. E., Foss-Feig, J. H., Malesa, E. E., Lee, E. B., Taylor, J. L., Newsom, C. R., ... Stone,
 W. L. (2011). Neurocognitive and behavioral outcomes of younger siblings of children with autism spectrum disorder at age five. *Journal of Autism and Developmental Disorders*, 42(3), 409–418. http://doi.org/10.1007/s10803-011-1263-4
- Wechsler, D. (2011). WASI-II: Wechsler abbreviated scale of intelligence. New York, NY: Psychological Corporation.
- White, S. W., Oswald, D., Ollendick, T., & Scahill, L. (2009). Anxiety in children and adolescents with autism spectrum disorders. *Clinical Psychology Review*, 29(3), 216–229. http://doi.org/10.1016/j.cpr.2009.01.003
- Wodka, E. L., Mathy, P., & Kalb, L. (2013). Predictors of phrase and fluent speech in children with autism and severe language delay. *Pediatrics*, 131(4), e1128–e1134. http://doi.org/10.1542/peds.2012-2221
- Woolfenden, S., Sarkozy, V., Ridley, G., & Williams, K. (2012). A systematic review of the diagnostic stability of autism spectrum disorder. *Research in Autism Spectrum Disorders*, 6(1), 345-354.
- World Health Organisation. (1993). Mental Disorders: A Glossary and Guide to their
 Classification in Accordance with the 10th Revision of the International Classification of
 Disease Research Diagnostic Criteria: ICD-10. Geneva: Author.

- Zwaigenbaum, L., Bryson, S. E., Brian, J., Smith, I. M., Roberts, W., Szatmari, P., ... &
 Vaillancourt, T. (2006). Stability of diagnostic assessment for autism spectrum disorder
 between 18 and 36 months in a high-risk cohort. *Autism Research*, 9(7), 790-800. doi: 10.1002/aur.1585.
- Zwaigenbaum, L., Bryson, S. E., Szatmari, P., Brian, J., Smith, I. M., Roberts, W., ... & Roncadin,
 C. (2012). Sex differences in children with autism spectrum disorder identified within a high-risk infant cohort. *Journal of autism and developmental disorders*, 42(12), 2585-2596.